Author's response to reviews

Title: Superparamagnetic iron oxide nanoparticles mediated 131I-hVEGF siRNA inhibits hepatocellular carcinoma tumor growth in nude mice

Authors:

wei xiong (xwbio2013@gmail.com)

Version: 2 Date: 27 March 2013

Author's response to reviews:

Dear Sir/Madam:
Thank you for your corrections/suggestions on our manuscript, the following are my responses:

1. We have a statement explaining our compliance with guidelines. "Their care, housing and all manipulations were conducted in accordance with the Guidelines for the Care and Use of Experimental Animals (Animal Care and Use Committee of Tongji Medical College, Wuhan, China). " This sentence can be found in "animal experiment" under "material and method" section.

2. We declared there is no conflict of interest. I put in after references, I have moved it to after main text.

3. I put the author's contribution section after references. I have moved it to after main text.

4. I'll add all author's info in the submission section

Thanks!

Best,

Xiong Wei